

Citation: Blake A, Hazel A, Jakurama J, Matundu J, Bharti N (2023) Disparities in mobile phone ownership reflect inequities in access to healthcare. PLOS Digit Health 2(7): e0000270. https://doi.org/ 10.1371/journal.pdig.0000270

Editor: Valentina Lichtner, University of Leeds, UNITED KINGDOM

Received: February 27, 2023

Accepted: May 5, 2023

Published: July 6, 2023

Copyright: © 2023 Blake et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The primary data used in this study are in Table A in <u>S1 Text</u>. All data and code necessary to reproduce the figures and analyses are publicly available: <u>https://github.com/</u> bhartilab/namibia_phone_bias.

Funding: This study was supported by The Branco Weiss – Society in Science fellowship (to NB) and The Huck Institutes of Life Sciences at Penn State University (to NB). Additional funding was provided by the joint NIH-NSF-NIFA Ecology and Evolution of Infectious Disease (award R01TW012434 to NB) and by the National Science Foundation (Grant No. **RESEARCH ARTICLE**

Disparities in mobile phone ownership reflect inequities in access to healthcare

Alexandre Blake¹, Ashley Hazel², John Jakurama³, Justy Matundu³, Nita Bharti¹*

1 Biology Department, Center for Infectious Disease Dynamics, Penn State University, University Park, Pennsylvania, United States of America, 2 Francis I. Proctor Foundation, University of California, San Francisco, California, United States of America, 3 Kaoko Information Center, Opuwo, Namibia

* nita@psu.edu

Abstract

Human movement and population connectivity inform infectious disease management. Remote data, particularly mobile phone usage data, are frequently used to track mobility in outbreak response efforts without measuring representation in target populations. Using a detailed interview instrument, we measure population representation in phone ownership, mobility, and access to healthcare in a highly mobile population with low access to health care in Namibia, a middle-income country. We find that 1) phone ownership is both low and biased by gender, 2) phone ownership is correlated with differences in mobility and access to healthcare, and 3) reception is spatially unequal and scarce in non-urban areas. We demonstrate that mobile phone data do not represent the populations and locations that most need public health improvements. Finally, we show that relying on these data to inform public health decisions can be harmful with the potential to magnify health inequities rather than reducing them. To reduce health inequities, it is critical to integrate multiple data streams with measured, non-overlapping biases to ensure data representativeness for vulnerable populations.

Author summary

Mobile phone data are increasingly used to inform public health efforts in both high and low-income settings due to convenience and growing phone penetration. However, digital inequities are ubiquitous and more pronounced in areas where mobile phone ownership is low or heterogeneous. The biases introduced by using mobile phone data to represent populations and their health care needs are rarely measured but have the potential to be detrimental to the most vulnerable segments of populations. We conducted detailed interviews measuring mobile phone ownership, mobility, and access to healthcare in mobile and remote populations in Namibia. We found that mobile phone owners represent a small proportion of the population that is highly mobile and has better access to healthcare. This is likely not unique. Due to the nature of their collection, mobile phone data often underrepresent vulnerable populations. This study demonstrates that uncritically using mobile phone data to inform public health decisions can perpetuate health inequities. 2202872 to NB). Funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Introduction Background

Human movement and contacts underlie pathogen transmission. Characterizing and quantifying movement greatly improves infectious disease surveillance, control, and prevention efforts. These efforts are particularly important to guide health policies for mobile populations [1-3]. The importance and challenge of measuring movement across spatiotemporal scales has given rise to a broad spectrum of methods to track human movements [4-7]. However, many methods strongly underrepresent or completely miss the most vulnerable populations, which are often most in need of improved public health services. This includes marginalized groups, low-income populations, small or low-density populations, and rural or remote populations.

Measuring movement

The past decade has seen growing use of novel data sources as proxies for human movement due to technological advances and convenience in the absence of readily available, representative data [8,9]. Current methods to quantify human movement include commercial air traffic [6], satellite derived anthropogenic illumination [5,10], and mobility traces derived from mobile phone call detail records (CDRs) within national boundaries [3,4]. Each of these approaches captures biased samples of populations, though the extent and types of biases vary. Using relatively new data streams to inform public health decisions can leave decision makers with unmeasured biases in population representation, which can harm efforts to improve health equity.

Justified by a broad temporal increase in the usership of mobile phones, CDRs in particular are increasingly used as proxies for human movement [9,11–13]. CDRs are collected for billing purposes from all mobile phones regardless of device capabilities, unlike GPS data or app-derived information. CDRs reflect mobile phone usage and document the towers that route each telecommunication transaction (call, text, or other billable event). Despite usership growth, mobile phone penetration is substantially lower in low-income countries when compared to high-income countries [14] and phone usage is heterogeneous in many low-income and under-resourced populations within middle- and high-income nations [15,16]. In 2022, phone ownership for individuals older than ten years was 92.9% in Europe and 88.5% in the Americas, and 60.6% in Africa [17]. Relentless advances in mobile communications technology have, in many ways, widened this gap and increased inequities that stem from access to technology. This is because technological advances have not improved access to phones but they have increased the advantages of phone ownership; this digital divide is associated with rurality, lower literacy levels, and lower gross domestic product at the national level [18], with additional subnational disparities [19,20].

Phone data analyses often incorrectly assume a 1:1 relationship between a person and a phone [21]. Some studies explore scalable solutions to correct for low ownership but cannot address biased usership [22]. The necessary anonymity of mobile device data renders it impossible to assess potential biases from the data itself (Fig 1A). This can result in unknowingly ignoring complex ownership and usership patterns. Additionally, the mobility patterns of mobile phone users cannot be tracked in areas that lack mobile phone reception, which is frequently the case in low-income, low population density areas. Overall, this leads to misrepresenting mobility patterns and underrepresenting the segments of populations most in need of improved health services. Public health efforts that are intended to reach the most vulnerable and least resourced members of populations that rely on biased data can result in interventions that are counterproductive to improving health equity. These populations are often small in size, which contributes to the reasons they are routinely overlooked. Regardless of their size,



Fig 1. Overview of concept and location. (A) Schematic showing underrepresentation and biased representation of a population. Circles represent individuals; black, blue, and green colors each represent one of three possible values for a demographic characteristic. Orange shaded areas represent individuals selected for inclusion in each data set. (B) Above: Inset map of continent, Namibia in black, and region of interest outlined in red. Below: Detailed map of the region of interest showing population density in 2016 (yellow = high; blue = low; white = NA (Etosha National Park, no human population)). Study area circled in orange, health clinics (orange crosses), regional hospital (red cross). Map created in ArcGIS with data from GADM (https://gadm.org/).

https://doi.org/10.1371/journal.pdig.0000270.g001

the most vulnerable populations define global health equity, and they must be prioritized instead of forgotten.

Phones and health in Namibia

Namibia is a middle-income nation with a relatively high infectious disease burden. The top ten causes of death include HIV/AIDS, lower respiratory infections, tuberculosis, and diar-rheal diseases [23]. Public health efforts in Namibia increasingly incorporate phone-derived data, specifically CDRs, to measure movement to understand spatial connectivity, trip duration, and seasonality to inform infectious disease prevention and management, including malaria elimination and HIV risk reduction [24–27]. In 2013, mobile phone ownership in Namibia was estimated at 95% in urban areas [28], with reliable network coverage. However, populations in Namibia vary greatly in size and density (Fig 1B). In Namibia's non-urban areas, where health improvements are needed most, phone ownership and network coverage had not been measured prior to this study. Additionally, mobile and remote populations are not consistently included in traditional censuses in the region [29,30].

The Kunene province is a desert area in northwestern region of Namibia (Fig 1B). Local residents are largely nomadic pastoralists who move seasonally and many are members of the Himba tribe. Residents travel primarily by walking. The area has a minimal, informal road network, and scarce access to petrol. There is a strong gender division of labor in these populations: men herd cattle and women manage childcare and subsistence farming. As a result, men travel greater distances and move more frequently than women do and men maintain control over valuable assets, specifically livestock. These nomadic populations represent vulnerable groups for both preventable and emerging infectious diseases. Although basic public healthcare is relatively affordable and accessible in urban areas of Namibia, in remote areas, facilities are scarce and distance to the nearest clinic is often prohibitive. In populations with limited access to healthcare, many illnesses go undiagnosed, untreated, and unreported [31].

To assess data representativeness in mobile phone-derived data in non-urban areas, we conducted detailed interviews among residents of the Kunene province (Fig 1B) in 2015 (rainy season) and 2016 (dry season). We collected data to compare characteristics of mobile phone owners and non-mobile phone owners among participants. Specifically, we assessed demographic characteristics (gender and age), self-reported recent movements and travel, and measures of access and barriers to healthcare (recency, frequency, distance, monetary cost, time cost) (see Supporting Information in S1 Text for survey details and Table O in S1 Text for survey instrument). We find that data derived from mobile phones in this remote population: 1) underrepresent women, 2) overestimate mobility and access to care, and 3) underrepresent remote and rural areas. Using mobile phone-based data to inform public health needs would largely overlook the segments of the population that need improvements the most. Instead, measuring biases in phone usage and integrating multiple data sources with non-overlapping biases can help include the most vulnerable members of populations in data that are used to inform public health efforts. This is an important step towards improving health equity, a priority for The United Nations' (UN) Sustainable Development Goals (SDG) [32].

Material and methods

Interviews

Sampling methods and recruitment. We conducted interviews in two settlements in the desert of Kaoko in the Kunene region, Namibia, in February of 2015 (rainy season) and October of 2016 (dry season) (Fig 1B). The vast majority of participants self-identified as members of the Himba tribe. Participants also included members of the Tjimba, Ovambo, Zemba, and Twe tribes. The Himba are the majority tribe in Kaoko, and all tribes in the area are Bantuspeaking and participate in the cattle-herding culture and economy. The lifestyle is highly mobile and semi-nomadic, with pastoralism supported by subsistence agriculture. We conducted interviews at the same two physical settlements across two years, and participants included residents of these settlements as well as visitors. These data represent mobile individuals and families from many locations and settlements within the Kaoko region of northern Namibia.

Interviews were performed in the local language, Otjiherero, with a translator who was the same gender as the participant.

Inclusion criteria. Study participants were restricted to adults. The designation of "adult" was locally determined by household responsibilities, interpersonal relationships (sexually active), or age, in instances when it was known. Locally, individuals are considered adults at approximately 16 years of age.

Pregnant women could participate in the study, as it posed no risk to the mother or fetus. Participants were required to provide consent to participate in the study and could revoke their consent at any point.

Sample size. We conducted a total of 167 interviews. We conducted interviews with 102 adults in 2015. Twenty-five of these (12 men, 13 women) were completed in a short format during a pilot phase during which we primarily collected demographic data. We conducted full-length interviews with 75 adults (37 men, 38 women). The remaining 2 adults (2 women) provided basic demographic information as child guardians. In 2016, we conducted full-length interviews with 65 adults (31 men, 34 women). Eight adults were interviewed in both 2015 and 2016.

Instrument items analyzed in this study. This study analyzes some of the data collected during the interviews. Included here are participants' answers to questions about phone

ownership, phone use during their lifetime, areas with phone reception, travel time to the nearest health center (discrete values in hours; estimated by participants in hours or based on sun positions), travel destinations in the previous 12 months (up to 5), mode of travel to the nearest health care center, and the ability to access health care when wanted. Participants were also asked about basic demographic information, individual and household resources, and sexual contacts in the previous 12 months (Table O in <u>S1 Text</u> contains the full survey instrument).

Statistical analysis

All analyses were done using R 4.0.3 [33]. We provide a basic description of the participants with means and percentages before multiple imputations (Table 1 and Table A in S1 Text). We addressed data missingness with multiple imputations (15 imputations) and analyzed every imputed data set before applying the Rubin rules to pool the estimates of interest [34,35] (see S1 Text). All the tests performed to compare groups (men vs women, participants interviewed in 2015 vs participants interviewed in 2016, or mobile phone owners vs non-owners) were performed using the imputed data sets. We calculated differences in means and in proportions in every imputed data set, pooled them together, and obtained p-values using Wald tests [35].

We used a principal component analysis to identify a reasonable proxy for access to health care. We applied it to a set of variables that addressed access to care and checked their loadings [36]. The set of variables included the number of travel destinations, travel to health care by car or another mode of transportation, the travel time to access healthcare, and the ability to access health care when desired. We used the loadings to identify the variable most strongly correlated with the main principal component and used that variable as a proxy for access to care in the rest of the analysis (see S1 Text).

We compared access to care between mobile phone owners and non-phone owners by calculating the mean values of the identified proxy for each group and pooling the differences in means of every imputed data set. To minimize confounding biases in this comparison, we calculated propensity score and calculated the difference in means after trimming and matching in every imputed data set (Figure B in <u>S1 Text</u>) before pooling the estimates to ensure that

| | Participants (N = 159) | | Mobile phone owners (N = 41) | | Non-phone owners (N = 91) | | NA for phone ownership (N = 27) | |
|--------|------------------------|------|---------------------------------|------|------------------------------|------|------------------------------------|------|
| | n | % | n | % | n | % | n | % |
| Year | | | | | | | | |
| 2015 | 102 | 64.2 | 26 | 63.4 | 49 | 53.8 | 27 | 100 |
| 2016 | 57 | 35.8 | 15 | 36.6 | 42 | 46.2 | 0 | 0 |
| Gender | | | | | | | | |
| Women | 85 | 53.5 | 11 | 26.8 | 59 | 64.8 | 15 | 55.6 |
| Men | 74 | 46.5 | 30 | 73.2 | 32 | 35.2 | 12 | 44.4 |
| Age | | | | | | | | |
| 16-25 | 36 | 22.6 | 10 | 24.4 | 23 | 25.3 | 3 | 11.1 |
| 26-35 | 41 | 25.8 | 18 | 43.9 | 22 | 24.2 | 1 | 3.7 |
| 36-45 | 31 | 19.5 | 6 | 14.6 | 18 | 19.8 | 7 | 25.9 |
| 46-59 | 24 | 15.1 | 5 | 12.2 | 12 | 13.2 | 7 | 25.9 |
| 60+ | 27 | 17.0 | 2 | 4.9 | 16 | 17.6 | 9 | 33.3 |

 Table 1. Summary of the characteristics of the participants including mobile phone ownership.
 Overview of participant characteristics before applying multiple imputations.

https://doi.org/10.1371/journal.pdig.0000270.t001

mobile phone owners and non-owners were as similar as possible [37] (see <u>S1 Text</u>, and Figure B in <u>S1 Text</u>).

We investigated the bias in access to care by estimating the distribution of the identified proxy among phone users only and for the whole sample by calculating the difference between the two groups. We calculated the probability mass for every value after applying Rubin rules to the imputed data sets by gender and mobile phone ownership and smoothed it with discrete kernel density estimators to minimize random noise [38,39]. We then estimated the distribution of the proxy among mobile phone owners as well as for the total population by calculating the weighted average of the smoothed distribution by gender and mobile phone ownership with pooled proportions from imputed data sets used as weights. Little to no difference between the two probability masses would point toward no bias. Conversely, a positive or a negative difference (with the distribution of mobile phone owners as a reference) point toward an over- or underrepresentation in mobile phone owners and for the total population after making single imputations embedded in bootstrap (1000 samples) with the percentile method [40].

We assessed the bias in movements that would be captured by mobile phone-based data by participant mobile phone ownership and by analyzing the distribution of the numbers of recently visited locations, including travel destinations (up to 6 reported per participant) and home locations. We classified all destinations according to mobile phone network reception availability based on our observations in the field with MTC mobile phones, the main service provider in Namibia, and participants' responses to questions about locations with and without phone reception. We classified destinations as follows: A) areas where phone reception was easily and widely available, B) areas with limited access to phone reception, including where reception was only available at elevation and required significant walking in mountainous terrain to reach, and therefore not accessible to everyone, C) areas with no access to mobile phone network reception, or D) areas with unlikely access to phone reception but for which we lacked definitive information on the presence or absence of reception. We calculated the percentage of participants' destinations and visits that would not be captured by mobile phone-based data.

Ethics

The study design has approval from Penn State's Institutional Review Board (IRB #STUDY00001510: Movement and Pathogens in Namibia) and Institutional Biosafety Committee (IBC #48898). Each interview began with an explanation of the survey process and purpose of the research. Data collection began after participants provided their formal verbal consent. Participants could decline to answer or skip any questions, decline continuation at any point during the survey, or revoke consent. Prior to conducting field work, the authors obtained research visas from the Namibian Ministry of Health and Social Services (MOHSS) and local institutional support through The University Center for Studies in Namibia (TUCSIN).

Results

Overview

Participant characteristics from interviews are presented in Table 1 and Table A in S1 Text. About two thirds of the total participants were interviewed in 2015 (102/167) and the remaining third were interviewed in 2016 (65/167). The gender ratio of participants was slightly in favor of women (80 men and 87 women, or 52.1% of women). About half of all participants were between 16 and 35 years old (49.1%), though participant age distribution varied by year;

23.5% vs 6.2% were aged 60+ years during 2015 and 2016, respectively. Out of all the participants who answered questions about mobile phone ownership, only 31.4% (44/140) of the participants reported owning a phone and 58.6% (82/140) reported that they had used a phone in their lifetime. 73.2% (101/138) of participants answered that they were unable to access a health care center when they wanted medical attention.

For the remaining results, we calculated three estimates due to the lack of independence of the data collected from the eight individuals who were interviewed in both 2015 and 2016: A) considering their responses in 2015 and 2016 as independent data points, B) excluding their data collected in 2015, and C) excluding their data collected in 2016. These estimates were very similar. When analyzing data from both collection years for the population, we include only the data collected in 2015 for these eight individuals. The resulting sample size is 159 unique participants, and 41 unique mobile phone owners (Table 1) (see <u>S1 Text</u> for full details and other estimates).

Mobile phone ownership was more frequent among participants interviewed in 2015 compared to 2016, 34.7% (26/75) vs 27.7% (18/65) (Table 1). The difference was not statistically different when tested on imputed datasets (p = 0.54). However, the proportion of participants who reported having ever used a mobile phone in their lifetime was lower in 2015 than in 2016; 48.0% (36/75) vs 70.8% (46/65) (Table A in S1 Text). This difference was statistically significant when tested on imputed datasets (p < 0.01). Overall, phone ownership and usership were significantly greater in men compared to women; 48.4% (30/62) of men vs 15.7% (11/70) of women reported phone ownership, and 75.8% (47/62) of men vs 42.9% (30/70) of women reported prior phone usership in their lifetime (Table 1 and Tables A and B in S1 Text). These differences were statistically significant when tested on imputed datasets (respectively p < 0.001for phone ownership and p < 0.001 for phone usership).

Phone owners traveled to a greater number of destinations than did non-phone owners in the 12 months preceding the interview, reporting means of 3.5 destinations vs 2.6 destinations (Fig 2). When pooling estimates on imputed datasets the mean number of destinations were.3.1 vs 2.1 respectively (p<0.01). When stratified by gender, this difference in travel history among phone owners was statistically significant in women (p<0.05) but not in men. Phone owners also experienced significantly shorter travel times to health care centers compared to non-phone owners, reporting means of 3.8 hours vs 5.7 hours in transit, respectively



Fig 2. Distributions of the number of travel destinations and travel time to healthcare for non-phone owners (blue) and mobile phone owners (red). The curves show density plots of the distributions of the number of travel destinations and travel time to healthcare (no data were imputed for these density plots), with a jittered rug below. The mean values are indicated by dashed vertical lines.

https://doi.org/10.1371/journal.pdig.0000270.g002

(p<0.05). When stratified by gender, the association between phone ownership and shorter travel times to health care centers remained significant in men (p<0.05) but not in women, among whom phone ownership was very low (Figures C and D and Tables C and D in <u>S1</u> Text).

To reduce the dimensionality of the data, we applied a principal component analysis (PCA) to all the previously specified variables that were related to access to health care. The travel time to a health care center was identified and then used as the main proxy for access to care for the remainder of the analysis (see <u>S1 Text</u>, Figure A in <u>S1 Text</u>).

Biased estimate of access to care

The smoothed distribution of travel time to healthcare highlighted a greater probability of shorter travel times among mobile phone owners, which was even more pronounced among men (Fig 3A). The average values for travel time to healthcare among mobile phone owners compared to the total population was 3.9 hours (95% CI: 3.0–5.8) vs 5.2 hours (95% CI: 4.4–7.0), respectively (Fig 3C). The smoothed distribution of travel time to healthcare taken from mobile phone owners would over-represent values below 5 hours and mostly under-represent values over 5 hours (Fig 3B and 3D).

The reduction in mean travel time after propensity score matching was consistent with greater access to care among phone owners and was close to significance, -2.6 hours of travel time to healthcare (95%CI: -5.5–0.3) (Tables I and J and Figure E in <u>S1 Text</u>).

Biased representation of mobility

Mobile phone derived data can only capture movements of mobile phone owners in areas with mobile phone network reception. The 41 participants who owned mobile phones (25.8% of 159 participants) reported 34.7% (186/536) of all 536 reported recent travel destinations (including their home settlements) (Fig 4).

Access to mobile phone reception did not differ significantly between the destinations of mobile phone owners and non-phone owners (Fig 4, Table L in S1 Text); 50.3% (257/511) of all visits were to areas with widely available network reception, 25.2% (129/511) of visits were to areas with limited access to network reception, 2.2% (11/511) of visits were to areas with no access to network reception. Although the access to network reception was similar between the destinations reported by mobile phone owners and non-phone owners, 73.8% (377/511) of the reported destinations would be missed by mobile phone-based data (Fig 4). Of the missed visits, 33.2% (125/377) were to areas with no access or unlikely but not confirmed access to mobile phone reception. Visits to areas with no phone reception, or unlikely but unconfirmed reception represented 24.5% (125/511) of all travel destinations reported by participants who answered questions related to mobile phone ownership.

Discussion

This study presents original, detailed data on a small number of people but comprehensively represents the populations interviewed. These data represent understudied populations that are often missed in routine data collection on demographics and health, have limited access to health care, and are not prioritized by current health policies. We minimized the effect of the small sample size by applying appropriate statistical approaches to avoid dropping data points and to minimize confounding factors.

With mobile phones being increasingly used in public health data collection [8,41,42], we measured biases in mobile phone-derived data that are often acknowledged, but rarely



Fig 3. Distributions of travel time to healthcare by gender and mobile phone ownership. (**A**) Distribution of travel time to healthcare by gender and mobile phone ownership (non-phone owners in blue and mobile phone owners in red) after applying a discrete kernel density estimator [38,39] on the distribution after multiple imputation (truncated for values above 15 hours). The dashed vertical lines show the average travel time to health care for each group after multiple imputations. (**B**) Distribution of travel time to healthcare considering only mobile phone owners (red) and total population (purple). Distributions are the results of a weighted average of the distribution in B using proportions of each category shown in A after multiple imputations. The dashed vertical population (purple) after multiple imputations. (**C**) Average travel time to healthcare for mobile phone owners (red) and for the total population (purple) after multiple imputations. (**C**) Average travel time to healthcare for mobile phone owners (red filled circle) and for the total population (purple filled circle) after multiple imputations and their 95% CI estimated by bootstrap. (**D**) Difference in the distribution of travel time to healthcare between mobile phone owners and the total population (the reference group). This is the difference of the two step lines displayed in B. For travel times to healthcare that reported more frequently by mobile phone owners than by the total population, the values are shown as positive. Light shades of grey represent smaller absolute differences.

https://doi.org/10.1371/journal.pdig.0000270.g003

quantified, and frequently ignored. Our analysis presents evidence that phone owners have better access to health care and greater mobility than members of the same population who do not own phones. We also show that mobile phone-based data provide a skewed perception of local mobility patterns. These data miss a majority of reported travel destinations due to low phone ownership. They also fail to include movement to areas without phone network reception, which tend to overlap with areas that are most in need of improved access to public health resources. Further, using mobile phone data to track the spread of a transmissible pathogen in a region like this one would miss most movements and contacts. This strategy would be highly misleading and ineffective for outbreak response efforts hoping to control the number of new cases and limit the spatial spread of a transmissible pathogen [21].

PLOS Digital Health | https://doi.org/10.1371/journal.pdig.0000270 July 6, 2023



Fig 4. Distribution of participant destinations by mobile phone network reception and mobile phone ownership. Each row represents a unique town and the length of each bar represents the number of participants who reported traveling there (no data were imputed for this histogram). Locations are divided vertically into categories based on mobile phone network reception. Visitor numbers are divided horizontally by visitors' mobile phone ownership (186 destinations from 41 mobile phone owners on the left, 325 destinations from 91 non-phone owners on the right, 25 destinations from the 27 participants with missing values for mobile phone ownership are not shown). Visits that would not be captured by mobile phone-based data are shown in dark grey, visits that would be captured by mobile phone-based data are shown in light grey.

https://doi.org/10.1371/journal.pdig.0000270.g004

Mobile phone ownership in remote populations of Himba pastoralists in Namibia was significantly more prevalent in men than women. Mobile phone owners reported much shorter travel times to health care centers compared to non-phone owners, as well as more time-efficient travel methods. Representing these populations with data collected from mobile phoneowners overestimates travel destinations per person and underestimates travel time to a health care center per person. Using mobile phone data to represent this population prevents the detection of the very inequities that require improvement. Failing to measure data biases while relying on biased data to guide policies reinforces existing health inequities.

Many low and middle-income countries (LMIC) report rapid growth in mobile phone usage and infrastructure [18,20]. It is often expected that this trend will continue globally and

that low or biased phone usership will simply be overcome with time. Our data suggest that universal growth of mobile phone ownership and infrastructure is unlikely; these interviews were conducted 19 months apart and phone ownership rates were constant and low, with a slight decrease over time. Mobile phone network coverage in the area also did not improve during this time. Subsequent visits to this area and these populations indicate that phone reception and ownership have not increased in recent years.

Although these populations are small in size and number, addressing health inequities for members of the Himba tribe and other underrepresented populations is necessary for public health progress. Remote populations often make up only a small percentage of national or regional populations, and working with these populations necessarily yields small sample sizes. However, ignoring small populations in public health efforts propagates cycles of underrepresentation in data. Remote populations play an important and often underappreciated role in the transmission, emergence, and persistence of infectious diseases [43,44]. Overlooking these groups hinders the final stages of elimination of vaccine preventable transmissible pathogens like polio and measles. In addition, a lack of access to basic health care in remote areas leads to delayed outbreak detection of endemic infections and emerging pathogens [45], which can increase morbidity and mortality. The UN emphasizes the global importance of equitable access to health care as a basic human right for all. Several of the SDGs prioritize improvements in health equity as critical milestones towards progress [32,46].

Large data sets cannot overcome inherent biases by virtue of their size alone [47]. Gender, wealth, education, and disability create gaps in phone ownership at various levels of aggregation across low and middle income countries (LMIC) [15,16,48,49]. These characteristics are also associated with differential access to healthcare [50,51]. Biases in population representativeness in CDRs are often estimated by comparison to another source in which data are aggregated spatially or reported at the population-level instead of the individual-level, such as income subgroups [52–54]. Unfortunately, it is very difficult to identify biases in data representativeness in aggregated data, because heterogeneities and inequities often intersect and are present at fine scales. Traditional data sources, such as population-based surveys, also contain biases in remote areas. When data collection methods have been in use for a long time, their biases become relatively well understood and, once detected, data collection efforts can be adjusted to reduce or account for biases in data representativeness. For example, this study collected data at the individual level. This level of granularity allowed us to measure biases while avoiding the uncertainty associated with comparison data that could be lost by aggregating across administrative areas, age groups, or income levels [52,53].

The Kunene region is not unique regarding remoteness, heterogeneous mobile phone ownership, and wireless signal coverage scarcity. Interviews that are specifically designed to assess populations in remote areas are labor, time, and cost intensive. However, they are critical to identify and monitor populations in need of public health improvements and to assess their representation in proxy measures, including mobile phone data. Surveys that collect data on both phone ownership and variables of interest in targeted areas make it possible to measure biases at comparable, operational scales. High granularity data collected from a source unrelated to mobile phones, such as surveys, are necessary to correct biased estimates of mobility from phones. Alternatively, some studies have successfully used GPS devices paired with surveys, predicated on an understanding of the acceptance and usefulness of GPS devices in a local context [55–58]. This approach is advantageous because it is not impacted by phone network reception or differential mobile phone ownership. Unfortunately, unpredictable longdistance travel over long time periods made GPS device retrieval impossible and we were unable to execute this strategy in northern Namibian pastoralist populations. Integrating multiple data sources improves estimates of the sizes and movements of mobile populations [59], improves data quality on health indicators, and can help guide decision makers towards reducing health inequalities. This strategy also augments the interpretability and usability of data streams like CDRs while minimizing their potential drawbacks.

Well-informed health policy decisions produce effective and equitable improvements in population health outcomes [60,61]. Prioritization is critical in resource-limited settings, as is impact assessment [62,63]. Barriers to implementing effective health policy include inaccurate and biased assessments of the underlying problems and the impacted populations [64]. Accurate estimates of the health needs of mobile or remote populations are critical for reducing the burden of infectious diseases and improving global health. Addressing problems of data representativeness is critical to ensure data sources do not become tools that enhance inequities.

Health care that serves all populations, no matter how small or remote, is integral to the management of infectious diseases. Pathogens persist in areas with insufficient disease surveillance and prevention [65]. Decisions based on data that have not been critically assessed for representativeness and biases can promote dangerous health inequities by incorrectly assuming inclusion of the least visible groups.

Supporting information

S1 Text. Supporting information providing additional details on the methods and results. (DOCX)

Acknowledgments

We thank Katuritara Tjiningire for providing valuable translation assistance during 2016 as well as logistical support and community relationship building. We thank Ephraim Hanks for providing advice and support on the statistical methods. We also thank Shweta Bansal for providing insightful comments and feedback on the manuscript. We are indebted to Namibian authorities, the Ministry of Health and Social Services, and all study participants for their time and participation, and for providing valuable information and assistance.

Author Contributions

Conceptualization: Nita Bharti.
Data curation: Ashley Hazel, John Jakurama, Justy Matundu, Nita Bharti.
Formal analysis: Alexandre Blake, Nita Bharti.
Funding acquisition: Nita Bharti.
Investigation: Alexandre Blake, Nita Bharti.
Methodology: Alexandre Blake, Nita Bharti.
Project administration: Ashley Hazel, John Jakurama, Justy Matundu, Nita Bharti.
Supervision: Nita Bharti.
Validation: Alexandre Blake.
Visualization: Alexandre Blake, Nita Bharti.
Writing – original draft: Alexandre Blake, Nita Bharti.
Writing – review & editing: Alexandre Blake, Ashley Hazel, John Jakurama, Justy Matundu, Nita Bharti.

References

- Camargo MCC, de Moraes JC, Souza VAUF, Matos MR, Pannuti CS. Predictors related to the occurrence of a measles epidemic in the city of São Paulo in 1997. Rev Panam Salud Pública. 2000 Jun; 7:359–65.
- Ruktanonchai NW, Bhavnani D, Sorichetta A, Bengtsson L, Carter KH, Córdoba RC, et al. Censusderived migration data as a tool for informing malaria elimination policy. Malar J [Internet]. 2016 [cited 2018 Aug 30];15. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC4864939/
- Tatem AJ, Qui Y, Smith DL, Sabot O, Ali AS, Moonen B. The use of mobile phone data for the estimation of the travel patterns and imported P. falciparum rates among Zanzibar residents. Malar J. 2009; 8 (1):12.
- 4. Gonzalez MC, Hidalgo CA, Barabasi AL. Understanding individual human mobility patterns. Nature. 2008 Jun 5; 453:779:782. https://doi.org/10.1038/nature06958 PMID: 18528393
- Bharti N, Tatem AJ, Ferrari MJ, Grais RF, Djibo A, Grenfell BT. Explaining seasonal fluctuations of measles in Niger using nighttime lights imagery. Science. 2011; 334:1424–7. <u>https://doi.org/10.1126/ science.1210554 PMID: 22158822</u>
- Colizza V, Barrat A, Barthelemy M, Vespignani A. The role of the airline transportation network in the prediction and predictability of global epidemics. Proc Natl Acad Sci USA. 2006; 103(7):2015:2020. https://doi.org/10.1073/pnas.0510525103 PMID: 16461461
- 7. Brockmann D, Hufnagel L, Geisel T. The scaling laws of human travel. Nature. 2006; 439(7075):462–5. https://doi.org/10.1038/nature04292 PMID: 16437114
- 8. Oliver N, Lepri B, Sterly H, Lambiotte R, Deletaille S, Nadai MD, et al. Mobile phone data for informing public health actions across the COVID-19 pandemic life cycle. Sci Adv [Internet]. 2020 Jun [cited 2022 Jan 17]; https://www.science.org/doi/abs/10.1126/sciadv.abc0764
- Bengtsson L, Lu X, Thorson A, Garfield R, von Schreeb J. Improved Response to Disasters and Outbreaks by Tracking Population Movements with Mobile Phone Network Data: A Post-Earthquake Geospatial Study in Haiti. Gething PW, editor. PLoS Med. 2011 Aug 30; 8(8):e1001083. https://doi.org/10. 1371/journal.pmed.1001083 PMID: 21918643
- Sutton P, Roberts D, Elvidge C, Baugh K. Census from Heaven: An estimate of the global human population using night-time satellite imagery. Int J Remote Sens. 2001 Jan 1; 22(16):3061–76.
- Lu X, Tan J, Cao Z, Xiong Y, Qin S, Wang T, et al. Mobile phone-based population flow data for the COVID-19 outbreak in mainland China. Health Data Sci. 2021; 2021. https://doi.org/10.34133/2021/ 9796431 PMID: 36405355
- Ruktanonchai NW, DeLeenheer P, Tatem AJ, Alegana VA, Caughlin TT, Erbach-Schoenberg E zu, et al. Identifying Malaria Transmission Foci for Elimination Using Human Mobility Data. PLOS Comput Biol. 2016 Apr 4; 12(4):e1004846. https://doi.org/10.1371/journal.pcbi.1004846 PMID: 27043913
- 13. Cumbane SP, Gidófalvi G. Spatial Distribution of Displaced Population Estimated Using Mobile Phone Data to Support Disaster Response Activities. ISPRS Int J Geo-Inf. 2021 Jun; 10(6):421.
- Ingram G. Bridging the global digital divide: A platform to advance digital development in low- and middle-income countries [Internet]. Brookings Institution Reports. Washington, United States: The Brookings Institution; 2021 Jun [cited 2023 Jan 27]. <u>https://www.proquest.com/docview/2576906035/</u> abstract/54AC8BB7A17B4C24PQ/1
- 15. Blumenstock JE, Eagle N. Divided We Call: Disparities in Access and Use of Mobile Phones in Rwanda. Inf Technol Int Dev. 2012 Jun 2; 8(2):1–16.
- Wesolowski A, Eagle N, Noor AM, Snow RW, Buckee CO. Heterogeneous Mobile Phone Ownership and Usage Patterns in Kenya. PLOS ONE. 2012 Apr 25; 7(4):e35319. https://doi.org/10.1371/journal. pone.0035319 PMID: 22558140
- ITU-D ICT Statistics [Internet]. International Telecommunication Union. [cited 2023 Feb 12]. <u>https://www.itu.int/itu-d/sites/statistics/</u>
- Billon M, Marco R, Lera-Lopez F. Disparities in ICT adoption: A multidimensional approach to study the cross-country digital divide. Telecommun Policy. 2009 Nov 1; 33(10):596–610.
- Blumenstock J, Eagle N. Mobile divides: gender, socioeconomic status, and mobile phone use in Rwanda. In: Proceedings of the 4th ACM/IEEE International Conference on Information and Communication Technologies and Development—ICTD '10 [Internet]. London, United Kingdom: ACM Press; 2010 [cited 2020 May 11]. p. 1–10. http://dl.acm.org/citation.cfm?doid=2369220.2369225
- **20.** Brännström I. Gender and digital divide 2000–2008 in two low-income economies in Sub-Saharan Africa: Kenya and Somalia in official statistics. Gov Inf Q. 2012 Jan 1; 29(1):60–7.
- Erikson SL. Cell Phones ≠ Self and Other Problems with Big Data Detection and Containment during Epidemics. Med Anthropol Q. 2018; 32(3):315–39.

- Lai S, Erbach-Schoenberg E zu, Pezzulo C, Ruktanonchai NW, Sorichetta A, Steele J, et al. Exploring the use of mobile phone data for national migration statistics. Palgrave Commun. 2019 Mar 26; 5(1):1– 10. https://doi.org/10.1057/s41599-019-0242-9 PMID: 31579302
- Vos T, Lim SS, Abbafati C, Abbas KM, Abbasi M, Abbasifard M, et al. Global burden of 369 diseases and injuries in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. The Lancet. 2020 Oct 17; 396(10258):1204–22. <u>https://doi.org/10.1016/S0140-6736(20)30925-9 PMID: 33069326</u>
- 24. Tatem AJ, Huang Z, Narib C, Kumar U, Kandula D, Pindolia DK, et al. Integrating rapid risk mapping and mobile phone call record data for strategic malaria elimination planning. Malar J. 2014; 13:52. https://doi.org/10.1186/1475-2875-13-52 PMID: 24512144
- Giles JR, Zu Erbach-Schoenberg E, Tatem AJ, Gardner L, Bjørnstad ON, Metcalf CJE, et al. The duration of travel impacts the spatial dynamics of infectious diseases. Proc Natl Acad Sci U S A. 2020 Aug 24; https://doi.org/10.1073/pnas.1922663117 PMID: 32839329
- Valdano E, Okano JT, Colizza V, Mitonga HK, Blower S. Using mobile phone data to reveal risk flow networks underlying the HIV epidemic in Namibia. Nat Commun. 2021 May 14; 12(1):2837. https://doi. org/10.1038/s41467-021-23051-w PMID: 33990578
- Wesolowski A, zu Erbach-Schoenberg E, Tatem AJ, Lourenço C, Viboud C, Charu V, et al. Multinational patterns of seasonal asymmetry in human movement influence infectious disease dynamics. Nat Commun. 2017 Dec 12; 8(1):2069. https://doi.org/10.1038/s41467-017-02064-4 PMID: 29234011
- Ministry of Health and Social Services—MoHSS/Namibia, ICF International. Namibia Demographic and Health Survey 2013 [Internet]. Windhoek, Namibia: MoHSS/Namibia and ICF International; 2014. http:// dhsprogram.com/pubs/pdf/FR298/FR298.pdf
- 29. Randall S. Where have all the nomads gone? Fifty years of statistical and demographic invisibilities of African mobile pastoralists. Pastoralism. 2015 Nov 4; 5(1):22.
- Abakar MF, Schelling E, Béchir M, Ngandolo BN, Pfister K, Alfaroukh IO, et al. Trends in health surveillance and joint service delivery for pastoralists in West and Central Africa. Future Pastor J Zinsstag E Schelling B Bonfoh Eds Rev Sci Tech Int Epiz. 2016; 35(2):683–91. https://doi.org/10.20506/rst.35.2. 2549 PMID: 27917961
- Hazel A, Ponnaluri-Wears S, Davis GS, Low BS, Foxman B. High prevalence of Neisseria gonorrhoeae in a remote, undertreated population of Namibian pastoralists. Epidemiol Infect. 2014 Nov; 142 (11):2422–32. https://doi.org/10.1017/S0950268813003488 PMID: 25267407
- 32. Kumar S, Kumar N, Vivekadhish S. Millennium Development Goals (MDGs) to Sustainable Development Goals (SDGs): Addressing Unfinished Agenda and Strengthening Sustainable Development and Partnership. Indian J Community Med Off Publ Indian Assoc Prev Soc Med. 2016; 41(1):1–4. <u>https://doi.org/10.4103/0970-0218.170955 PMID: 26917865</u>
- R Core Team. R: A language and environment for statistical computing [Internet]. Vienna, Austria: R Foundation for Statistical Computing; 2022. URL https://www.R-project.org/
- Barnard J, Rubin DB. Small-Sample Degrees of Freedom with Multiple Imputation. Biometrika. 1999; 86(4):948–55.
- Marshall A, Altman DG, Holder RL, Royston P. Combining estimates of interest in prognostic modelling studies after multiple imputation: current practice and guidelines. BMC Med Res Methodol. 2009 Jul 28; 9:57. https://doi.org/10.1186/1471-2288-9-57 PMID: 19638200
- 36. Abdi H, Williams LJ. Principal component analysis. WIREs Comput Stat. 2010; 2(4):433–59.
- Rubin DB. Estimating causal effects from large data sets using propensity scores. Ann Intern Med. 1997 Oct 15; 127(8 Pt 2):757–63. https://doi.org/10.7326/0003-4819-127-8_part_2-199710151-00064 PMID: 9382394
- Nagler T. A generic approach to nonparametric function estimation with mixed data [Internet]. arXiv; 2018 [cited 2022 Aug 22]. http://arxiv.org/abs/1704.07457
- **39.** Nagler T. Asymptotic analysis of the jittering kernel density estimator [Internet]. arXiv; 2017 [cited 2022 Aug 22]. http://arxiv.org/abs/1705.05431
- Brand J, van Buuren S, le Cessie S, van den Hout W. Combining multiple imputation and bootstrap in the analysis of cost-effectiveness trial data. Stat Med. 2019; 38(2):210–20. https://doi.org/10.1002/sim. 7956 PMID: 30207407
- Ashigbie PG, Rockers PC, Laing RO, Cabral HJ, Onyango MA, Mboya J, et al. Phone-based monitoring to evaluate health policy and program implementation in Kenya. Health Policy Plan. 2021 May 17; 36 (4):444–53. https://doi.org/10.1093/heapol/czab029 PMID: 33724372
- Denkinger CM, Grenier J, Stratis AK, Akkihal A, Pant-Pai N, Pai M. Mobile health to improve tuberculosis care and control: a call worth making. Int J Tuberc Lung Dis Off J Int Union Tuberc Lung Dis. 2013 Jun; 17(6):719–27.

- Feldmann H, Czub M, Jones S, Dick D, Garbutt M, Grolla A, et al. Emerging and re-emerging infectious diseases. Med Microbiol Immunol (Berl). 2002 Oct 1; 191(2):63–74. <u>https://doi.org/10.1007/s00430-002-0122-5 PMID: 12410344</u>
- Sbarra AN, Rolfe S, Nguyen JQ, Earl L, Galles NC, Marks A, et al. Mapping routine measles vaccination in low- and middle-income countries. Nature. 2021 Jan; 589(7842):415–9. https://doi.org/10.1038/ s41586-020-03043-4 PMID: 33328634
- Matson MJ, Chertow DS, Munster VJ. Delayed recognition of Ebola virus disease is associated with longer and larger outbreaks. Emerg Microbes Infect. 2020 Jan 1; 9(1):291–301. <u>https://doi.org/10.1080/</u> 22221751.2020.1722036 PMID: 32013784
- **46.** Pandey S. The Road From Millennium Development Goals to Sustainable Development Goals by 2030: Social Work's Role in Empowering Women and Girls. Affilia. 2017 May 1; 32(2):125–32.
- Bradley VC, Kuriwaki S, Isakov M, Sejdinovic D, Meng XL, Flaxman S. Unrepresentative big surveys significantly overestimated US vaccine uptake. Nature. 2021 Dec; 600(7890):695–700. <u>https://doi.org/ 10.1038/s41586-021-04198-4</u> PMID: 34880504
- Gillwald A, Mothobi O. After Access 2018: A demand-side view of mobile Internet from 10 African countries. 2019;
- 49. Aranda-Jan C, Tech GA, Nique M, Tech GA, Pitcher S, Tech GA, et al. The Mobile Disability Gap Report 2020. London: GSMA. Mob Disabil Gap Rep. 2020; 4:4.
- Pennington A, Orton L, Nayak S, Ring A, Petticrew M, Sowden A, et al. The health impacts of women's low control in their living environment: A theory-based systematic review of observational studies in societies with profound gender discrimination. Health Place. 2018 May 1; 51:1–10. <u>https://doi.org/10.1016/j.healthplace.2018.02.001</u> PMID: 29482064
- Seidu AA, Darteh EKM, Agbaglo E, Dadzie LK, Ahinkorah BO, Ameyaw EK, et al. Barriers to accessing healthcare among women in Ghana: a multilevel modelling. BMC Public Health. 2020 Dec 17; 20 (1):1916. https://doi.org/10.1186/s12889-020-10017-8 PMID: 33334326
- Wesolowski A, Eagle N, Noor AM, Snow RW, Buckee CO. The impact of biases in mobile phone ownership on estimates of human mobility. J R Soc Interface. 2013 Apr 6; 10(81):20120986. <u>https://doi.org/ 10.1098/rsif.2012.0986 PMID: 23389897</u>
- Pestre G, Letouzé E, Zagheni E. The ABCDE of Big Data: Assessing Biases in Call-Detail Records for Development Estimates. World Bank Econ Rev. 2020 Feb 1; 34(Supplement_1):S89–97.
- Meppelink J, Van Langen J, Siebes A, Spruit M. Beware Thy Bias: Scaling Mobile Phone Data to Measure Traffic Intensities. Sustainability. 2020 Jan; 12(9):3631.
- 55. Paz-Soldan VA, Stoddard ST, Vazquez-Prokopec G, Morrison AC, Elder JP, Kitron U, et al. Assessing and Maximizing the Acceptability of Global Positioning System Device Use for Studying the Role of Human Movement in Dengue Virus Transmission in Iquitos, Peru. Am J Trop Med Hyg. 2010 Apr; 82 (4):723–30. https://doi.org/10.4269/ajtmh.2010.09-0496 PMID: 20348526
- 56. Paz-Soldan VA, R CR Jr, Morrison AC, Stoddard ST, Kitron U, Scott TW, et al. Strengths and Weaknesses of Global Positioning System (GPS) Data-Loggers and Semi-structured Interviews for Capturing Fine-scale Human Mobility: Findings from Iquitos, Peru. PLoS Negl Trop Dis. 2014 Jun 12; 8(6):e2888. https://doi.org/10.1371/journal.pntd.0002888 PMID: 24922530
- Vazquez-Prokopec GM, Stoddard ST, Paz-Soldan V, Morrison AC, Elder JP, Kochel TJ, et al. Usefulness of commercially available GPS data-loggers for tracking human movement and exposure to dengue virus. Int J Health Geogr. 2009 Nov 30; 8(1):68. https://doi.org/10.1186/1476-072X-8-68 PMID: 19948034
- Krenn PJ, Titze S, Oja P, Jones A, Ogilvie D. Use of Global Positioning Systems to Study Physical Activity and the Environment: A Systematic Review. Am J Prev Med. 2011 Nov 1; 41(5):508–15. https://doi.org/10.1016/j.amepre.2011.06.046 PMID: 22011423
- 59. Wild H, Glowacki L, Maples S, Mejía-Guevara I, Krystosik A, Bonds MH, et al. Making Pastoralists Count: Geospatial Methods for the Health Surveillance of Nomadic Populations. Am J Trop Med Hyg. 2019; 101(3):661–9. https://doi.org/10.4269/ajtmh.18-1009 PMID: 31436151
- Wright J, Williams R, Wilkinson JR. Development and importance of health needs assessment. BMJ. 1998 Apr 25; 316(7140):1310–3. https://doi.org/10.1136/bmj.316.7140.1310 PMID: 9554906
- 61. Biomedicine I of M (US) C on the S and EI of D in, Bulger RE, Bobby EM, Fineberg HV. The Formulation of Health Policy by the Three Branches of Government [Internet]. Society's Choices: Social and Ethical Decision Making in Biomedicine. National Academies Press (US); 1995 [cited 2020 May 19]. <u>https://www.ncbi.nlm.nih.gov/books/NBK231979/</u>
- Hanney SR, Gonzalez-Block MA, Buxton MJ, Kogan M. The utilisation of health research in policy-making: concepts, examples and methods of assessment. Health Res Policy Syst. 2003 Jan 13; 1(1):2. https://doi.org/10.1186/1478-4505-1-2 PMID: 12646071

- 63. Tang JL, Griffiths S. Review paper: epidemiology, evidence-based medicine, and public health. Asia Pac J Public Health. 2009 Jul; 21(3):244–51. <u>https://doi.org/10.1177/1010539509335516</u> PMID: 19443874
- Oliver K, Lorenc T, Tinkler J, Bonell C. Understanding the unintended consequences of public health policies: the views of policymakers and evaluators. BMC Public Health. 2019 Aug 6; 19(1):1057. https://doi.org/10.1186/s12889-019-7389-6 PMID: 31387560
- 65. Klepac P, Metcalf CJE, McLean AR, Hampson K. Towards the endgame and beyond: complexities and challenges for the elimination of infectious diseases. Philos Trans R Soc B Biol Sci. 2013 Aug 5; 368 (1623):20120137. https://doi.org/10.1098/rstb.2012.0137 PMID: 23798686